Author's response to reviews

Title: Response rates and selection problems, with emphasis on mental health variables and DNA sampling, in large population-based, cross-sectional and longitudinal studies of adolescents in Norway

Authors:

Espen Bjertness (espen.bjertness@medisin.uio.no)
Aase Sagatun (aase.sagatun@r-bup.no)
Kristian Green (kristian.green@medisin.uio.no)
Lars Lien (lars.lien@medisin.uio.no)
Anne Johanne Søgaard (anne.johanne.sogaard@fhi.no)
Randi Selmer (randi.selmer@fhi.no)

Version: 8 Date: 12 October 2010

Author's response to reviews:

I have done two changes in the manuscript:

1) Page 20, paragraph 1, line: 5: "although" is substituted with "but".

2) Acknowledgements on page 26 is changed to:
   The data collection was carried out and funded by the University of Oslo and the Norwegian Institute of Public Health. The City of Oslo contributed to funding of the baseline study while The Centre for Child and Adolescent Mental Health, Eastern and Southern Norway, Oslo, contributed to funding of the follow-up study.